

# Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation

Nikolina Jovanović<sup>(0,1,2\*</sup> Žan Lep<sup>(0,3,4</sup> Jelena Janković<sup>(0,5</sup> Aysegul Dirik<sup>(0,2,6</sup> Anna Rees<sup>(0,5</sup> and Maev Conneely<sup>(0,2,6,7)</sup>

<sup>1</sup>Centre for Psychiatry and Mental Health, Wolfson Institute of Population Sciences, Queen Mary, University of London, London, UK

<sup>2</sup>East London NHS Foundation Trust, London, UK

<sup>3</sup>Department of Psychology, Faculty of Arts, University of Ljubljana, Ljubljana, Slovenia

<sup>4</sup>Centre for Applied Epistemology, Educational Research Institute, Ljubljana, Slovenia

<sup>5</sup>Community Perinatal Service, Birmingham and Solihull Mental Health Foundation Trust, London, UK

<sup>6</sup>Unit for Social and Community Psychiatry (WHO Collaborating Centre for Mental Health Service Development), Queen Mary University of London, London, UK

<sup>7</sup>Division of Psychiatry, University College London, London, UK

\*Corresponding author n.jovanovic@qmul.ac.uk

Published October 2024 DOI: 10.3310/YTRK6337

#### Abstract

**Background:** During pregnancy and the postpartum period, women's mental health can deteriorate quickly. Timely and easy access to services is critically important; however, little is known about the pathways women take to access services. Previous research has shown that women from ethnic minority groups in the United Kingdom experience more access issues compared to the White British women.

**Aim:** To describe pathways taken to specialist community perinatal mental health services and explore how they vary across services and ethnic groups.

**Methods:** This is a two-site, longitudinal retrospective service evaluation conducted in Birmingham and London during 6 months (1 July–31 December 2019). Electronic records of 228 women were accessed and data were extracted on help-seeking behaviour, referral process and the type of pathway (i.e. simple or complex). Data were collected using the adapted World Health Organization encounter form and analysed using uni- and multivariable analyses.

**Results:** The median time from the start of perinatal mental illness to contact with perinatal mental health services was 20 weeks. The majority of patients accessed perinatal mental health services through primary care (69%) and their pathway was simple, that is they saw one service before perinatal mental health services (63%). The simple pathway was used as a proxy for accessible services. In Birmingham, compared to London, more referrals came from secondary care, more women were experiencing current deterioration in mental health, and more women followed a complex pathway. Despite differences between ethnic groups regarding type of pathway and duration of patient journey, there was no evidence of difference when models controlled for confounders such as clinical presentation, general characteristics and location. The service's location was the strongest predictor of the type of pathway and duration of patient journey.

**Limitations:** The heterogeneity among categorised ethnic groups; data extracted from available electronic records and not validated with patient's own accounts of their pathways to care; unanalysed declined referrals; the study was conducted before the COVID-19 pandemic and pathways may be different in the post-COVID-19 period.

**Conclusion:** The study provides important insights into how patients find their way to community perinatal mental health services. It shows that there is a great degree of variability in the time taken to get into these services, and the pathway taken. This variation does not come from different needs of patients or different clinical presentations but rather from service-level factors.

This article should be referenced as follows:

Jovanović N, Lep Z, Janković J, Dirik A, Rees A, Conneely M. Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation. [published online ahead of print October 30 2024]. Health Soc Care Deliv Res 2024. https://doi.org/10.3310/YTRK6337

**Future work:** The studied community perinatal mental health services in the United Kingdom operate with a significant degree of variability in the types and characteristics of patient pathways. Future research should explore these issues on the national and international levels. Additionally, future research should explore the reasons for the different pathways taken and the outcomes and risks associated with them.

**Funding:** This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Health and Social Care Delivery Research programme as award number 17/105/14.

A plain language summary of this research article is available on the NIHR Journals Library Website https://doi.org/10.3310/YTRK6337.

#### Introduction

Mental health disorders frequently manifest during perinatal period, which encompasses pregnancy and the first year after childbirth.<sup>1,2</sup> These disorders entail particular risks due to the potential rapid deterioration of a woman's mental health, which can significantly impact both her pregnancy and the well-being of the infant.<sup>3</sup> Failure to effectively treat perinatal mental health disorders results in substantial health, societal and economic burdens.<sup>4</sup> Prompt and convenient access to mental health services is of paramount importance for the well-being of women and their families.

In the United Kingdom (UK), specialised perinatal mental health services (PMHS) have been established to focus on assessing, diagnosing and treating women aged 18 years and older who experience moderate to severe mental health disorders during the preconception, antenatal and postnatal periods. These services treat women presenting with antenatal and postnatal depression, anxiety disorders, eating disorders, relapses of severe mental illnesses and postpartum psychosis. These disorders can develop gradually or suddenly during pregnancy and after childbirth. Many patients also report problems with emotional bonding with their baby, substance misuse, exposure to domestic violence, and social and financial difficulties. Consequently, the risk of abuse or neglect towards the baby can be increased, often necessitating a multiagency approach and the involvement of child social services.

Since 2016, significant investments have been made into PMHS to ensure women with moderate to severe mental health issues receive prompt, evidence-based treatments.<sup>5</sup> Extensive efforts have provided guidelines on the essential components and care pathways for developing and delivering these services.<sup>6-8</sup> Many women have reported positive experiences with PMHS, often due to strong patient–clinician relationships, consistent care, and dedicated, non-judgmental clinicians.<sup>9,10</sup> However, significant issues related to access and engagement have also been reported. These include a lack of awareness of available services and mental illness, and unresponsiveness from PMHS.<sup>10,11</sup>

Furthermore, several major evidence gaps have been identified in the delivery of care within perinatal mental health care.<sup>6</sup> These include diversity and the inclusion of women from ethnic minority communities, LGBTQ+ [lesbian, gay, bisexual, transgender,queer (or sometimes questioning), intersex, asexual, and others] parents and fathers during the perinatal period. In our previous research, we explored access and utilisation of services by pregnant and postnatal women from ethnic minority communities. A population-based study conducted in England in 2017 revealed that Black African, Asian (all subgroups), and White Other (excluding English, Welsh, Scottish, Northern Irish, British, Irish or Romani ethnic groups) women exhibited lower utilisation of community mental health services and higher rates of inpatient admissions compared to White British women.<sup>12</sup> These findings suggest that the issues are primarily related to access rather than reflecting different levels of need.

This study was designed to explore patient pathways to community PMHS in the UK, including variations in pathways between services and among ethnic groups. This is the first study to explore patient pathways to PMHS. The concept of pathway-to-care studies developed by the World Health Organization (WHO) was used. Pathways-to-care studies represent a valid and cost-effective tool to provide information about patient access to psychiatric care.<sup>13,14</sup>

#### Methods

#### Study design

This is a two-site, longitudinal retrospective service evaluation study. It is reported in line with the Strengthening the Reporting of Observational Studies in Epidemiology checklist for observational studies.

#### Data collection

Data were collected using the adapted WHO encounter form.<sup>13</sup> The original questionnaire is a standardised schedule for gathering basic sociodemographic, clinical and pathways data for each participant. The questionnaire was adapted to perinatal services by the research team, piloted by an independent researcher and refined based on feedback received from a panel of researchers based at the Queen Mary University of London, and a Lived Experience Advisory Panel assembled for the project. The adaptation ensured that specific aspects of perinatal care pathways were captured, for example 'encountered services' included midwifery teams, obstetric services, children's social services, etc. (see *Report Supplementary Material 1*). The Lived Experience Advisory Panel met regularly throughout the project and advised on all aspects of the study, including interpretation of findings and dissemination.

#### Settings and patient eligibility

The study was conducted in community PMHS in Birmingham (Birmingham and Solihull Mental Health NHS Foundation Trust) and London (East London NHS Foundation Trust) during a 6-month period (July-December 2019).

Eligible patients were newly referred to services at least once during this time period. Patients who had received treatment from PMHS in the last 2 years prior to this were excluded because their past experience could have impacted the pathway they took and so biased the results. As mentioned, these services are commissioned to care for women presenting with moderate to severe mental illness. The referrals that did not meet the PMHS threshold for moderate to severe mental illness were declined and these were not included in the study. Only accepted referrals were analysed. Declined referrals were often redirected for support within primary care and thirdsector organisations. The list of patients was obtained from information technology services in participating mental health trusts. Clinicians accessed patient electronic records to extract data on patients' sociodemographics, clinical characteristics, the involvement of children's social care and pathways to PMHS.

#### **Studied variables**

The adapted WHO encounter form gathered information on each clinician/service encountered on the care pathway; the duration of the patient's journey to the service; the source of referral; reason for referral; and treatment offered. Based on the collected data, the following variables were created: help seeking initiated by patient or by others, the referrer to PMHS [primary care (e.g. family doctor)/secondary care (e.g. specialist medical services)], main reason for referral to PMHS (current deterioration of mental health/review of pre-existing condition); number of clinicians encountered on pathway to PMHS; type of pathway ('simple' - contact with one clinician/services before contact with PMHS or 'complex' - contacts with two or more clinicians/ services before PMHS); treatment offered (monitoring and support; medication; counselling; combined treatment); outcome of first perinatal appointment (accepted under PMHS/discharged); and time (in weeks) between the

start of perinatal mental health problems (self-reported or diagnosed) and the first appointment with PMHS. If a patient saw the same clinician or service twice or more in a month, particularly if this was due to rescheduling appointments, it was noted as one contact.

#### Data analysis

Descriptive statistics were used to report study variables. Mean (M), standard deviation (SD), median (Mdn), range and frequencies were used as appropriate. Ethnicity was initially collected for 18 categories taken from the British census and classified into five groups (White, Mixed, Asian, Black and other), as this facilitates comparison with public sector documents that also recommend the approach.<sup>15,16</sup> Due to small number of patients in ethnic groups 'Mixed' (n = 3) and 'other' (n = 4), data were reported (see *Appendix 1*) but they were excluded from quantitative analyses comparing the groups and these analyses were performed on three groups only (Asian, Black and White).

The normality of the distribution of study variables was tested using Shapiro–Wilk test. Comparisons were done using chi-square, independent sample *t*-tests, Mann–Whitney and Kruskal–Wallis tests as appropriate (see the notes in *Tables 1–3*). In cases where using a chi-squared test was illadvised due to small cell counts, Fisher's exact test for count data was used and compared with Monte Carlo simulated (100,000 replications) chi-squared test which allowed for the calculation of Bonferroni-corrected post hoc tests. Where Kruskal–Wallis test suggested significant differences in rank across groups, post hoc pairwise Wilcox tests with Benjamini and Hochberg (1995) adjustment to control for false discovery rate were used to compare ethnic groups.<sup>18</sup>

For testing the association of various predictors with simple care pathway to PMHS, a binary logistic regression was performed. Two sets of predictors were selected based on clinical relevance and results from univariate analyses (i.e. 'Patient-related variables' and 'Pathway-related variables'). They were entered in the regression model at the same time (*Table 4*). A 5% alpha error was used as the limit of statistical significance for each predictor.

Linear regression models testing the association of various independent variables and the amount of time between symptoms developing and access to PMHS were performed as shown in *Table 5*. We first established that there was no significant multicollinearity between the predictors [all variance inflation factors (VIFs) were between 1.02 and 1.46], and then entered all the predictors into the model at the same time.

Data analysis was carried out using SPSS [(version 22.0), IBM Corp (2013). IBM SPSS Statistics for Macintosh, Version

22.0. Armonk, NY: IBM Corp) and R (The R Foundation for Statistical Computing, Vienna, Austria, R Core Team R, R: A language and environment for statistical computing, 2013) using base [(version 4.2.2), stats (version 4.2.2), R Core Team (2022). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria], psych [(version 2.2.9), William Revelle (2022). psych: Procedures for Psychological, Psychometric, and Personality Research. Northwestern University, Evanston, IL, R package version 2.2.9, https://CRAN.R-project.org/ package=psych Revelle: not a software but an author of psych, DescTools [(version 0.99.47), Signorell A (2022). DescTools: Tools for descriptive statistics. R package version 0.99. 47. https://github.com/AndriSignorell/ DescTools/. CRAN. 2022], rcompanion [(version 2.4.18), Mangiafico Salvatore S (2022). rcompanion: Functions to support extension education program evaluation. version 2.4.18 Rutgers Cooperative Extension. New Brunswick, NJ. https://CRAN.R-project.org/package=rcompanion] and chisq.posthoc.test (version 0.1.2, Ebbert D (2019). Chisq. posthoc. test: a post hoc analysis for Pearson's chi-squared test for count data. R package version 0.1.2) packages.

#### Results

### Participating services and established referral pathways

Please refer to *Table 1* for the key characteristics of the services included in the study. Both services included in the study cater to ethnically diverse communities in innercity areas. The Birmingham service operates at a higher capacity, which reflects the larger number of live births in its catchment area. In 2019, the number of live births in the studied Birmingham catchment area (n = 8000) was nearly double that of the London one (n = 4331). The comparison of the ethnic composition of study participants with the maternity populations in Birmingham and London showed no difference (ps > 0.05; see *Table 1*). This indicates that, in terms of ethnicity, patients under the care of PMHS in the present study were similar to those in the general maternity population.

In both services, established referral pathways included primary care services (e.g. general practitioners, community midwifes, health visitors and counsellors) and secondary care services (e.g. community mental health teams, inpatient psychiatric wards, crisis resolution and home treatment teams, specialist midwives, obstetricians and emergency services). The crisis resolution and home treatment teams offer assessment and treatment for individuals experiencing a mental health crisis, as an alternative to hospital admission. It is worth noting that specialist midwives are part of primary care pathway for mental health and part of secondary care for maternity services. At the time of the study, self-referrals were not possible. Both services did not accept direct referrals from social care; instead, clinicians advised that the patient is first seen within healthcare services and then referred to PMHS.

#### Study sample

The study sample is described in *Table 2*. The study analysed records of 228 patients treated in Birmingham (n = 131, 58%) and London (n = 97, 42%). The majority of patients were pregnant (n = 166, 73%), while others were in the postnatal period (n = 54, 23%) or referred for preconception review (n = 8, 4%). Study participants in Birmingham, compared to London, were statistically significantly younger (p = 0.003), more women were pregnant (p = 0.008), and from Asian ethnicities (p = 0.007). There were few White British women in the Birmingham sample (p < 0.001) and overall Asian women were more likely to be unemployed (p < 0.001).

### Pathways to community perinatal mental health services

Detailed information about pathways to PMHS can be found in *Table 3*. The majority of patients (n = 146, 64%)initiated seeking help for their perinatal mental health problems. The initial contacts were largely within primary care (n = 157, 69%). In more than half of participants (n = 150, 65.8%), the first clinician/service that was contacted made a referral to PMHS. Approximately half of participants (n = 121, 53%) were referred to PMHS by primary care clinicians. The vast majority of participants (n = 195, 86%) were referred because their mental health deteriorated and required intervention from mental health services. Others were referred for reviewing and monitoring of a pre-existing condition (n = 33, 14%). In Birmingham, referrals to PMHS came more often from secondary care (p = 0.001). In London, review and monitoring of pre-existing conditions was a more common reason for referral (p = 0.002). After the first PMHS assessment, patients were offered monitoring and support (56%), medication (22%), counselling (13.6%) and combined treatment (8.3%). Seeing a larger number of carers on the pathway to PMHS was associated with a longer period between the first referral and the first perinatal appointment (r = 0.51, p < 0.001).

### Differences in pathways between ethnic groups

The included variables and how they varied across the five ethnic groups can be seen in *Tables 2* and *3*. The study results revealed that several parameters varied among ethnic groups.

#### TABLE 1 Characteristics of included community services at the time of study (2019)

	Birmingham (East team)	London (Tower Hamlets Perinatal Mental Health Service)
Location	Inner city	Inner city
Facility	Service is based both in community mental health base and in maternity hospital antenatal department.	Service is based within a general hospital.
Live births	8003	4331
Ethnicity (alphabetica	l order)	
Ethnicity UK 2021 Census dataª	Asian, Asian British or Asian Welsh 31.0%; Black, Black British, Black Welsh, Caribbean or African 10.9%; Mixed or multiple ethnic groups 4.8; other ethnic group 4.5%; White 48.6%.	Asian, Asian British or Asian Welsh 44.4%; Black, Black British, Black Welsh, Caribbean or African 7.3%; Mixed or multiple ethnic groups 5%; other ethnic group 3.9%; White 39.4%.
Ethnicity maternity data <sup>b</sup>	<ul> <li>National maternity data set (2019–20):</li> <li>Birmingham Women's hospital: Asian/Asian British 29%; Black 4%; other 5%; White 46%.</li> <li>University Hospital Birmingham: Asian/Asian British 25%; Black 4%; other 5%; White 51%.</li> </ul>	North East London STP (2019–20): Asian 36%; Black 11%; Mixed 8%; other 3%; White 41%; unknown 1%.
Ethnicity of study participants	Asian 32.1%; Black 4.6%; Mixed 1.5%; other 0%; White 61.8%.	Asian 46.4%; Black 9.3%; Mixed 4.1%; other 4.1%; White 30.9%; missing data 5.2%.
Staff		
Clinical staff <sup>c</sup>	12.8 FTE staff: 1.2 FTE consultant psychiatrist, 1 FTE team manager, 0.8 FTE clinical psychologist, 1 FTE assistant psychologist, 2 FTE advanced nurse practitioners, 3 FTE mental health practitioners, 0.8 FTE social worker, 2 FTE nursery nurse, 1 FTE occupational therapist.	8.2 FTE staff: 1.2 FTE consultant psychiatrist, 0.8 FTE team manager, 1 FTE clinical psychologist, 1 FTE trainee psychologist, 3 FTE specialist nurses, 0.2 FTE mental health specialist midwife, 1 FTE nursery nurse.
Admin staff	2	1
Service accreditation <sup>d</sup>	No	No
Self-referrals	Not possible	Not possible

a UK 2021 Census data reported for the whole Birmingham and for the London borough of Tower Hamlets.

b Data for Birmingham are sourced from the National Maternity data set, while data for London are sourced from Hospital Episode Statistics. STP stands for Sustainability and Transformation Plan. Patients under the care of the Birmingham PMHS are admitted to two specified hospitals: Birmingham Women's Hospital and University Hospital Birmingham.

c FTE stands for 'full-time equivalent' used for staff in full-time employment.

d Accreditation is conducted through The Perinatal Quality Network (PQN) which started accreditation and peer appraisal of community services in 2013. Accredited services must meet 100% type 1 standards, at least 80% type 2 standards and 60% type 3 standards.<sup>17</sup>
 Note

The differences in ethnic composition between the maternity population and the present study sample were not significant overall: Birmingham women's hospital ( $\chi^2$  = 3.03, *p* = 0.39), University Hospital Birmingham ( $\chi^2$  = 2.64, *p* = 0.45), NE London STP ( $\chi^2$  = 4.21, *p* = 0.22).

Compared to Asian and Black patients, White patients initiated help seeking more often (p = 0.040) and they were more likely to have complex care pathways (p = 0.046).

### Simple and complex pathways to perinatal mental health services

The total number of clinicians/services encountered on the pathway to PMHS ranged from 2 to 5 (Mdn 2). The majority of participants in the study went through so-called 'simple pathways' (n = 144, 63.2%). The number of complex care pathways cases was higher in Birmingham than in London (p < 0.001) (see Table 3).

Looking into the factors influencing the likelihood of having a simple care pathway to PHMS, the odds for such pathway were higher for patients in London, while other predictors were not significant when controlling for the shared variance (see *Table 4*). Regarding ethnicity, there was no evidence of difference when the model controlled for other potentially confounding variables (i.e.

This article should be referenced as follows:

Jovanović N, Lep Z, Janković J, Dirik A, Rees A, Conneely M. Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation. [published online ahead of print October 30 2024]. Health Soc Care Deliv Res 2024. https://doi.org/10.3310/YTRK6337

#### TABLE 2 Study sample

		Participating se	rvices		Ethnicity data <sup>a</sup>						
Study variables	All n = 228	Birmingham n = 131 (57.5%)	London n = 97 (42.5%)	Statistics	Asian n = 77 (33.8%)	Black n = 15 (6.6%)	White n = 111 (48.7%)	Statistics			
Age, mean (SD)	30.3 (5.7)	29.3 (5.2)	31.5 (5.9)	t = 2.994, df = 226, p = 0.003	30.3 (5.5)	32.0 (5.7)	30.0 (5.7)	F(1,201) = 0.22, p = 639			
Perinatal status, n (%)				p = 0.003 ( $\chi^2 = 11.11$ , p = 0.003)				p = 0.276			
Pregnant	166 (72.8)	106 (80.9)	60 (61.9)		56 (72.7)	12 (80.0)	83 (74.8)				
Postnatal	54 (23.7)	23 (17.6)	31 (32.0)		17 (22.1)	2 (13.3)	27 (24.3)				
Preconception	8 (3.5)	2 (1.5)	6 (6.2)		4 (5.2)	1 (6.7)	1 (0.9)				
Marital status, n (%)				χ <sup>2</sup> = 4.108, df = 1, p = 0.043				<i>p</i> = 0.460			
Single/divorced/living alone	36 (15.8)	26 (19.8)	10 (10.3)		11 (14.3)	4 (26.7)	18 (16.2)				
Married/cohabitating/ in relationship	189 (82.9)	102 (77.9)	87 (89.7)		65 (84.4)	11 (73.3)	92 (82.9)				
Education, n (%)				χ <sup>2</sup> = 0.280, df = 1, p = 0.597				<i>p</i> = 0.400			
Primary + secondary	74 (32.5)	42 (32.1)	32 (33.0)		24 (31.2)	3 (20.0)	43 (38.7)				
College + University (highest achieved)	108 (47.4)	57 (43.5)	51 (52.6)		35 (45.5)	9 (60.0)	52 (46.8)				
Employment, <i>n</i> (%)				χ <sup>2</sup> = 0.583, df = 1, p = 0.445				p < 0.001 (χ <sup>2</sup> = 17.96, p < 0.001) <sup>b</sup>			
Employed	97 (42.5)	58 (44.3)	39 (40.2)		18 (23.4)	9 (60.0)	58 (52.3)				
Unemployed	119 (52.2)	65 (49.6)	54 (55.7)		55 (71.4)	5 (33.3)	49 (44.1)				
Number of children, Mdn (minimum-maxi- mum, IQR)	Mdn = 1 (0-6, 2)	Mdn = 1 (0-6, 2)	Mdn = 1 (0-5, 2)	U = 6029, p = 0.630	Mdn = 1 (0−5, 2.75)	Mdn = 1 (0-4, 2)	Mdn = 1 (0-6), 2	$\chi^2(2) = 2.95,$ p < 0.229°			
Involvement of chil- dren's social services, n (%)				$\chi^2 = 0.145,$ df = 1, p = 0.703				p = 0.495			
Yes	81 (35.5)	48 (36.6)	33 (34.0)		26 (33.8)	4 (26.7)	44 (39.6)				
No	143 (62.7)	81 (61.8)	62 (63.9)		51 (66.2)	11 (73.3)	65 (58.6)				

a Data on ethnicity were missing for 25 women (10.9%). Due to missing values, some percentages may not add up to 100%. Due to small sample sizes and non-normal distribution of ordinal/interval variables, the significance of differences in access to PMHS across ethnic groups were calculated only for Asian, Black and White patients using.

b Fisher's exact.

c Kruskal–Wallis tests. With significant Fisher's exact tests, Monte Carlo simulated (100,000 replications) chi-squared test results are presented in parentheses, which allowed for the calculation of post hoc tests of significance. Mixed and other group were excluded from the analyses due to small sample size, but counts are presented informatively.

patient- and pathway-related variables). Still, the overall model was better at predicting the type of pathway compared to model with no predictors. The complex pathways are presented in detail in *Appendix 2*. In Birmingham, the most frequent complex pathway, present in 29% of patients, was to access the

#### TABLE 3 Key aspects of pathways to PMHS

		Participating servi	ces		Ethnicity data					
Study variables	All, n = 228	Birmingham n = 131 (57.5%)	London n = 97 (42.5%)	Statistics	Asian n = 77 (33.8%)	Black n = 15 (6.6%)	White n = 111 (48.7%)	Statistics		
Help-seeking initiated by, <i>n</i> (%)ª				$\chi^2(1) = 1.385, p = 0.284$				<i>p</i> = 0.005 (χ <sup>2</sup> = 10.13, <i>p</i> = 0.009)		
Patient	145 (63.6)	98 (74.8)	48 (49.5)		43 (55.8)	8 (53.3)	81 (73.0)			
Others	17 (7.5)	9 (6.9)	8 (8.2)		11 (14.3)	1 (6.7)	3 (2.7)			
Referrer to perinatal, n (%)º				$p = 0.002 (\chi^2 = 12.6, p = 0.001)$				p = 0.510 <sup>b</sup>		
Primary care	123 (53.9)	59 (45.0)	62 (63.9)		46 (59.7)	9 (60.0)	58 (52.3)			
Secondary care	104 (45.6)	72 (55.0)	35 (36.1)		31 (40.3)	5 (33.3)	53 (47.7)			
Main reason for referral, <i>n</i> (%)				$\chi^2(1) = 9.185, p = 0.002$				p = 0.438 <sup>b</sup>		
Current deterioration of mental health	195 (85.5)	120 (91.6)	75 (77.3)		62 (80.5)	13 (86.7)	97 (87.4)			
Review of pre-existing condition	33 (14.5)	11 (8.4)	22 (22.7)		15 (19.5)	2 (13.3)	14 (12.6)			
Number of professionals/ services encountered before PMHS, mean (SD)	M = 2.5 (0.7), range 2-5	M = 2.7 (0.8)	M = 2.2 (0.5)	U = 3.823, p < 0.001	M = 2.3 (0.6)	M = 2.4 (0.8)	M = 2.6 (0.8)	χ <sup>2</sup> = 6.72, df = 2, <i>p</i> = 0.035 <sup>d</sup>		
Type of pathway, <i>n</i> (%) <sup>e</sup>				$\chi^2(1) = 36.434,$ p < 0.001				$p = 0.030 \ (\chi^2 = 7.13, p = 0.027)^{b}$		
Simple	144 (63.2)	61 (46.6)	83 (85.6)		57 (74.0)	11 (73.3)	62 (55.9)			
	84 (36.8)	70 (53.4)	14 (14.4)		20 (26.0)	4 (26.7)	49 (44.1)			

This article should be referenced as follows: Jovanović N, Lep Z, Janković J, Dirik A, Rees A, Conneely M. Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation. [published online ahead of print October 30 2024]. Health Soc Care Deliv Res 2024. https://doi.org/10.3310/YTRK6337

#### TABLE 3 Key aspects of pathways to PMHS (continued)

		Participating servic	es		Ethnicity data			
Study variables Treatment, n (%) Monitoring and support Medication Counselling Combined treatment Outcome of first PMHS appointment, n (%) Accepted under PMHS	All, n = 228	Birmingham n = 131 (57.5%)	London n = 97 (42.5%)	Statistics	Asian n = 77 (33.8%)	Black n = 15 (6.6%)	White n = 111 (48.7%)	Statistics
Treatment, n (%)				$\chi^2(3) = 4.54, p = 0.209$				p = 0.452 <sup>b</sup>
•	128 (56.1)	68 (51.9)	60 (61.9)		44 (57.1)	9 (60.0)	60 (54.1)	
Medication	50 (21.9)	29 (22.1)	21 (21.6)		19 (24.7)	2 (13.3)	24 (21.6)	
Counselling	31 (13.6)	23 (17.6)	8 (8.2)		11 (14.3)	1 (6.7)	17 (15.3)	
	19 (8.3)	11 (8.4)	8 (8.2)		3 (3.9)	3 (20.0)	10 (9.0)	
first PMHS appointment,				$\chi^2(1) = 14.065,$ p < 0.001				χ² = 5.01, df = 2, p = 0.082
	151 (66.2)	100 (76.3)	51 (52.6)		49 (63.6)	7 (46.7)	81 (73.0)	
Discharged	77 (33.8)	31 (23.7)	46 (47.4)		28 (36.4)	8 (53.3)	30 (27.0)	
Time between the start of perinatal mental health problems and contact with PMHS (weeks) <sup>f</sup>	Mdn = 20.4 (0-276)	Mdn = 54.0 (0-276)	Mdn = 5.2 (0-236)	U = 1356, p < 0.001	Mdn = 13.0 (0-236)	Mdn = 7.4 (0.7-192)	Mdn = 48.0 (0-228)	χ <sup>2</sup> = 4.02, df = 2, <i>p</i> = 0.134 <sup>d</sup>

a Missing data N = 15 patients.

b Fisher's exact.

c N = 1 patient. Primary care included general practitioners, community midwives, health visitors, talking therapies or counsellors; secondary care included Community Mental Health Team (CMHT), crisis resolution and home treatment team, obstetricians, accident and emergency, etc.

d Kruskal-Wallis tests.

e Type of pathway is defined as 'simple' – contacts with one clinician/services before any contact with PMHS or 'complex' – contacts with two or more clinicians/services before PMHS. f N = 68 patients.

Notes

Statistics: Due to small sample sizes and non-normal distribution of ordinal/interval variables, the significance of differences in access to PMHS across ethnic groups was calculated only for Asian, Black and White patients using chi-square.

With significant Fisher's exact tests, Monte Carlo simulated (100,000 replications) chi-squared test results are presented in parentheses, which allowed for the calculation of post hoc tests of significance. Mixed and other group were excluded from the analyses due to small sample size, but counts are presented informatively.

ω

#### TABLE 4 Results of binary logistic regression for experiencing simple pathway to PMHS

	Predictors	Odds ratio	SE(b)	Wald z	p
Pathway-related variables	Intercept	36.43	2.79	1.29	0.198
	Location (London vs. Birmingham)	7.17	0.57	3.49	≤ 0.001
	Help seeking initiated by patient	7.88	1.18	1.75	0.079
	Main reason for referral - current deterioration of mental health	0.27	0.79	-1.64	0.104
Patient-related variables	Age (years)	0.96	0.05	-0.88	0.381
	Education	0.58	0.48	-1.12	0.264
	Employment	1.30	0.51	0.52	0.605
	Ethnicity (Asian vs. White)	0.70	0.54	-0.65	0.515
	Ethnicity (Black vs. White)	3.93	1.33	1.03	0.304
	Number of children	0.95	0.19	-0.28	0.779
	Children's social services are involved with the family	2.52	0.55	1.67	0.095

Note

The sample was reduced due to missing values (n = 113), improvement over the null model was significant [ $\chi^2(11) = 175.94$ , p < 0.001], pseudo  $R^2$ s: McFadden = 0.40, Nagelkerke = 0.57.

TABLE 5 Linear regression models for the time between the start of perinatal mental health problems until access to PMHS (in weeks)

Predictor	В	SE(b)	Т	p
Intercept	-0.62	46.61	-0.01	0.989
Service (Birmingham vs. London)	51.14	10.05	5.09	< 0.001
Simple care pathway (vs. complex)	-8.77	9.83	-0.89	0.374
Ethnicity (Asian vs. White)	-0.35	9.45	-0.04	0.971
Ethnicity (Black vs. White)	13.60	17.26	0.79	0.432
Age (years)	1.45	0.85	1.70	0.092
Involvement of children's social services (yes)	-5.61	9.05	-0.62	0.537
Main reason for referral – current deterioration of mental health	-32.14	13.20	-2.44	0.016

Notes

All VIFs were between 1.02 and 1.46, model fit: F(7,134) = 6.10, p < 0.001,  $R^2 = 0.24$ .

Two sets of predictors selected based on clinical relevance and results from univariable analyses (i.e. 'patient-related variables' and 'pathway-related variables') were entered in the regression model at the same time. A 5% alpha error was used as the limit of statistical significance for each predictor.

PMHS through both community midwife and a specialist midwife. In London, the most frequent complex pathway included crisis services such as accident and emergency (A&E) and crisis resolution and home treatment teams (36%). Six patients encountered five carers on their journey, five in Birmingham and one in London. Five of these women were pregnant and experiencing current deterioration in mental health. One woman was in the postnatal period, also deteriorating. Two of these women were suicidal and they utilised the crisis pathway.

## Duration of patient journey to perinatal mental health services

The study also explored duration of time in weeks between the self-reported or diagnosed start of perinatal mental health problems and first appointment with PMHS. Large variation was found and the range was from < 1 week to 276 weeks; the median was 20.4 weeks or 4.5 months. *Table 5* shows the linear regression model testing the association between the time between the start of perinatal mental health problems and the first appointment with PMHS based on the service location, the experience of a simple care pathway, ethnicity, age, involvement of social services and main reason for referral. The time until the first perinatal appointment with PMHS in weeks was shorter for patients in the London subsample (p < 0.001), and those who were experiencing current deterioration in mental health (p = 0.016). The results suggest that ethnicity was not a significant predictor of the time to first perinatal appointment. Taken together, around a quarter of the variance in time from experiencing symptoms to being seen by PMHS can be explained by the variables included in the model ( $R^2 = 0.24$ ).

#### Discussion

#### Main findings

- Primary care services emerged as the primary point of contact for most patients seeking help in the perinatal period, highlighting the crucial role of general healthcare providers and midwives in recognising perinatal mental health issues. The majority of patients accessed PMHS through primary care and their pathway was simple, that is they saw one service before PMHS.
- Several issues related to access to PMHS were identified. There was a pronounced variation in time between first experiencing symptoms and obtaining help from services, there were significant differences in pathways between the participating services, and approximately one-third of the sample came into PMHS via a so-called 'complex' pathway.
- The study did not reveal substantial differences in the variables studied between different ethnic groups, indicating a degree of equality in access to PMHS.
- The study highlights the influence of service location on patient pathways, encouraging further exploration of this phenomenon to better tailor mental health services to patients' needs.
- The majority of study participants sought help for their mental health problems, thus indicating that opening services to self-referrals could further improve access to PMHS.

#### Strengths and limitations

The study has several strengths. This is the first ever study to explore care pathways to PMHS in the UK and globally. The study offered unique insights into the type and length of patient journeys to PMHS. The issue of accessibility of PMHS for women from ethnic minority groups had previously not been explored and results from this study can be used when improving access to PMHS. The study adapted the WHO encounter form, which can be used by other services and researchers in the future. The study also has several limitations. Due to pragmatic reasons, ethnic groups were simplified into merged groups (e.g. Black, Asian, White), which can be problematic because this obscures any within-group differences. For example, within the Asian group, Pakistani and Bangladeshi have higher levels of diagnosed ill health than the White population in England.<sup>19</sup> Furthermore, it has been well recognised that these large ethnic groups are distinguished by their cultural, linguistic, religious and economic variations. Next, the sample size was limited, thus preventing exploration of pathways across more specific ethnic groups. The study samples matched the population of the Birmingham and East London catchment areas in terms of ethnicity, indicating that the results are generalisable to these areas. However, it is important to note that results may not be generalisable to other areas with different ethnic compositions. Data were extracted from available clinical records and not validated with patient's own accounts of their pathways to care. In future studies, it is advisable to conduct interviews with patients to ensure comprehensive documentation of all interactions with services, as not all of these may be recorded in electronic records. The quality of data depends on the quality of available records. The study focused on accepted referrals and declined referrals were not analysed, although this could have provided additional information about service access. However, the primary reason for declining referrals was not meeting the commissioned services' threshold, which requires individuals to have moderate to severe mental illness. Next, inconsistencies between records across services could have contributed to identified differences in pathways. The study was conducted before the COVID-19 pandemic, so pathways may be different in the post-COVID-19 period mainly due to poorer access to primary care, further expansion and increased capacity and visibility of PMHS. The first encountered service or professional could have been outside healthcare services (e.g. community group or religious leaders), which was not sufficiently explored. Self-referrals were not available at the time of the study; most services started offering them in 2021-2. This could change pathways, especially since we found that most women initiated contact with services; however, currently, services still do not receive extensive number of self-referrals (e.g. eight patients in the London service during 2022). This study focused only on patients who were referred to PMHS for the first time. It may be that there are entirely different patterns of pathway use for women who are being referred back into PMHS. The studied pathways were categorised into simple versus complex. This can be considered a simplification, as the type of pathway is defined by the size of the patientservice contacts, not their quality.

#### Interpretation of study findings

As mentioned, patients' initial contacts with services were primarily within so-called universal services, such as primary care, maternity services and health visiting. Notably, in over half of cases, the first clinician or service contacted for women's perinatal mental health disorder made the referral to PMHS. This type of pathway, dubbed a 'simple pathway', is suggestive of good accessibility, an effective referral system and good responsiveness from services. As mentioned earlier, untreated perinatal mental health disorders can pose substantial risks to both the pregnancy and the infant. Hence, having access to services through a straightforward pathway can offer significant advantages. These findings also provide evidence for effective partnerships within the wider system of care working with women and families, as recommended by the NHS England guidance on pathways in perinatal mental health care.<sup>8</sup> They also confirm that the perinatal period offers a unique window of opportunity to detect and treat mental illness, due to increased frequency of contact with healthcare professionals during regular antenatal and postnatal visits.<sup>2</sup>

The study also revealed a few worrying findings in regard to access to PMHS. There was a pronounced variation in time between first experiencing symptoms and obtaining help from services, with median time being 20 weeks. The prolonged duration of patients' journey to PMHS is a concern, as it may result in substantial risks to both the pregnancy and the infant due to delays in care delivery. It is important to note that women with delayed access to services could have received support from family, friends or community organisations.<sup>20</sup> However, we can argue that adding specialist treatment to existing support could facilitate recovery and certainly be helpful for women with little external support.

This study also showed that White women, compared to Asian and Black, more frequently went through complex pathways to access PMHS. However, in regression models, ethnicity was not a variable that accounted for a significant variance in the type of pathway or time taken to get into PMHS. This can be considered as a positive finding; however, more research into this matter is needed and studies with a larger sample size and with less merging of ethnic groups can provide further insights. Previous research has indicated that services need to make more efforts to address this issue and ensure that women from under-represented groups are able to access any support needed.<sup>21-23</sup>

The study also found significant differences in pathways between participating services. In Birmingham, patients were more often referred to PMHS from secondary care services, they were more often experiencing deterioration in mental health, and they more often went through complex pathways. The most frequent complex pathway was characterised by more contacts with midwives and this can be explained by the fact that the Birmingham subsample included more pregnant women. Additionally, inclusion of specialist midwifes, which then made the pathway more complex, was not necessarily a 'negative' finding. This may have lowered barriers for midwives and contributed to more pregnant women being referred to PMHS. In London, patients referred to PMHS were predominately coming from primary care, the main reason for referral was reviewing and monitoring of pre-existing condition, and patients more often went through simple pathways. The most frequent complex pathway can be described as crisis pathway and included presentation to A&E and crisis resolution and home treatment teams.

The role of service location and accompanying factors requires further exploration. The variation identified in this study was not related to differences in clinical needs between patients, indicating it is other, possibly spurious factors playing a part in shaping women's experience of service. In-depth research should explore the factors driving this. Different methods will be needed to obtain a full picture of what is happening: including service-level data, and perhaps ethnographic research to observe and understand how referrals happen (or do not happen) and the decision-making processes involved. These findings and future research should also consider the wider literature and recent reports which point to different groups having different experiences in services, and the role that might play in determining pathways to care.<sup>21,24</sup>

### Implications for clinical practice and research

The study demonstrates the benefits of investigating patients' referral pathways, rather than focusing solely on the caseload/access statistics. Services could use the adapted WHO encounter form for regular audits of a sample of patients accessing PMHS as an overall quality measure of the efficiencies of their pathways across services.

Furthermore, healthcare service standards appear to play a crucial role in determining patient pathways to PMHS. Improving service-level standards is essential for enhancing patient outcomes, satisfaction and overall healthcare system performance. To achieve this, a comprehensive and systematic approach is needed, which starts with conducting in-depth research and data analysis to identify gaps and areas of improvement. Future research should explore service-level factors that have

This article should be referenced as follows:

Jovanović N, Lep Z, Janković J, Dirik A, Rees A, Conneely M. Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation. [published online ahead of print October 30 2024]. Health Soc Care Deliv Res 2024. https://doi.org/10.3310/YTRK6337

the potential to affect the accessibility of services and patient pathways (e.g. partnerships with key stakeholders, allocation of resources, and support for patients regarding transportation and child care). Additionally, research should investigate the outcomes and risks associated with different pathways.

There is a remarkable dearth of research on the association between pathways to and through care and outcomes. This study suggests there is an enormous degree of variation. It should be considered that an assumption has been made in this study that more complex journeys with more specific healthcare clinicians is worse than simple and shorter journeys. Qualitative studies of women experiencing PMHS support the notion that a complex pathway, involving encounters with numerous services and carers, can be disruptive.<sup>9,10</sup> Women often describe feeling abandoned and as if they had to fight for help, which does not make them feel cared for by the services.9 Future studies could also explore whether simple pathways were actually associated with better health and well-being outcomes (also based on the background variables explored in this study) - in other words, what is the contribution of the pathway above and beyond those variables.

An important implication for care that arises from this study's findings is the reminder to not assume all women have the same experience. Clinicians and everyone encountered on women's pathways should realise the importance that each contact has, and that the clinician may not have good understanding of what has come before them on a patient's journey. It seems beneficial to discuss this point during patient's first contact with the service.

#### Equality, diversity and inclusion

This research focused on understanding the inequalities and inequities that persist in perinatal mental health settings. Equality, diversity and inclusion is an integral aspect of this work, as evidenced by our research team's diverse experiences related to perinatal mental illness, ethnic background, migration history and professional disciplines (clinical psychiatry, psychology and research roles). Additionally, we carefully considered language pertaining to gender and ethnicity. The participant populations were diverse and inclusive in terms of ethnicity and migration status. For this study, data were extracted from the records of patients under PMHS. We were not able to verify how they describe or identify themselves. We also acknowledge that combining ethnic groups is not ideal, as discussed under Limitations. We made efforts to report data for all ethnic groups, including 'Mixed' and 'other' that were not used in the main analysis due to small numbers.

#### Conclusion

This is the first ever study to explore patient pathways to community PMHS. The results reveal the intricate connections among the various parts of the mental health care system. The study shows that there was a great degree of variability in the time taken to get into community PMHS, and the pathway taken. It appears that this variation does not come from different needs of patients or different clinical presentations but rather from service-level factors. Future research should explore reasons for different pathways taken, and outcomes and risks associated with different pathways.

#### Additional information

#### **CRediT contribution statement**

**Nikolina Jovanović (https://orcid.org/0000-0002-7554-9837):** Conceptualisation (lead), Investigation (lead), Methodology (lead), Formal analysis (supporting), Supervision (lead), Writing – original draft (lead), Writing – editing and reviewing (lead).

Zan Lep (https://orcid.org/0000-0003-0130-4543): Formal analysis (lead), Writing – original draft (supporting), Writing – editing and reviewing (supporting).

**Jelena Janković (https://orcid.org/0000-0001-5542-0051):** Conceptualisation (supporting), Supervision (equal), Writing – editing and reviewing (supporting).

AysegulDirik(https://orcid.org/0000-0003-0812-8458):Datacuration(lead), Project administration(lead),Supervision(supporting), Writing - editing and reviewing(supporting).

Anna Rees (https://orcid.org/0009-0001-1855-9270): Data curation (equal), Writing – editing and reviewing (supporting).

**Maev Conneely** (https://orcid.org/0000-0001-8326-4498): Writing – original draft (supporting), Writing – editing and reviewing (supporting).

#### **Ethics statement**

The study is part of the National Institute for Health and Care Research-funded (NIHR) project entitled 'Accessibility and acceptability of perinatal mental health services for women from Ethnic Minority groups (PAAM)' (17/105/14). This study used service-level data. As there was no contact with patients for the study, Health Research Authority approval was not required, and obtaining consent from individuals was not required. The two participating services provided local approvals.

12

#### Information governance statement

East London NHS Foundation Trust (ELFT) is committed to handling all personal information in line with the UK Data Protection Act (2018) and the General Data Protection Regulation (EU GDPR) 2016/679. Under Data Protection legislation ELFT is the Data Processor; the Department for Health and Social Care (DHSC) is the Data Controller, and we process personal data in accordance with their instructions. You can find out more about how we handle personal data, including how to exercise your individual rights and the contact details for DHSC's Data Protection Officer https:// www.nihr.ac.uk/documents/nihr-privacy-policy/12242#howwe-protect-your-personal-data.

#### **Disclosure of interests**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/YTRK6337.

Primary conflicts of interest: None.

#### Data-sharing statement

All data requests should be submitted to the corresponding author for consideration. Access to anonymised data may be granted following review.

### Department of Health and Social Care disclaimer

This publication presents independent research commissioned by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, MRC, NIHR Coordinating Centre, the Health and Social Care Delivery Research programme or the Department of Health and Social Care.

This article was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

#### **Study registration**

The study was conducted as part of the larger NIHR-funded research project, which received approval from the Health Research Authority through the Research Ethics Committee (REC reference: 19LO1830, 264632).

#### Funding

This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Health

and Social Care Delivery Research programme as award number 17/105/14.

This article reports on one component of the research award Accessibility and acceptability of perinatal mental health services for women from Ethnic Minority groups (PAAM). For more information about this research please view the award page (https://www. fundingawards.nihr.ac.uk/award/17/105/14).

#### About this article

The contractual start date for this research was in May 2019. This article began editorial review in October 2023 and was accepted for publication in June 2024. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The Health and Social Care Delivery Research editors and publisher have tried to ensure the accuracy of the authors' article and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

#### Copyright

Copyright © 2024 Jovanović *et al.* This work was produced by Jovanović *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

#### List of abbreviations

A&E	accident and emergency
PMHS	perinatal mental health services
WHO	World Health Organization

#### List of supplementary material

**Report Supplementary Material 1** Adapted perinatal WHO encounter form

Supplementary material can be found on the NIHR Journals Library report page (https://doi. org/10.3310/YTRK6337).

This article should be referenced as follows:

Jovanović N, Lep Z, Janković J, Dirik A, Rees A, Conneely M. Pathways to specialist community perinatal mental health services: a two-site longitudinal retrospective service evaluation. [published online ahead of print October 30 2024]. Health Soc Care Deliv Res 2024. https://doi.org/10.3310/YTRK6337

Supplementary material has been provided by the authors to support the report and any files provided at submission will have been seen by peer reviewers, but not extensively reviewed. Any supplementary material provided at a later stage in the process may not have been peer reviewed.

#### References

- 1. NHS England. *Perinatal Mental Health*. URL: www. england.nhs.uk/mental-health/perinatal/ (accessed 13 April 2024).
- Howard LM, Khalifeh H. Perinatal mental health: a review of progress and challenges. World Psychiatry 2020;19:313–27.
- 3. Knight M. The findings of the MBRRACE-UK confidential enquiry into maternal deaths and morbidity. *Obstet Gynaecol Reprod Med* 2019;**29**:21–3.
- Bauer A, Knapp M, Parsonage M. Lifetime costs of perinatal anxiety and depression. J Affect Disord 2016;**192**:83–90.
- 5. The Mental Health Taskforce. The Five Year Forward View for Mental Health. 2016. URL: www. england.nhs.uk/wp-content/uploads/2016/02/ Mental-Health-Taskforce-FYFV-final.pdf (accessed 13 April 2024).
- Royal College of Psychiatrists. College Report CR232: Perinatal Mental Health Services: Recommendations for the Provision of Services for Childbearing Women. 2021. URL: www.rcpsych.ac.uk/docs/default-source/ improving-care/better-mh-policy/college-reports/ college-report-cr232---perinatal-mental-heath-services.pdf?Status=Master&sfvrsn=82b10d7e\_4 (accessed 13 April 2024)
- 7. Pan London Perinatal Mental Health Network. *Perinatal Mental Health Care Pathways October* 2015. 2015. URL: www.england.nhs.uk/london/wp-content/uploads/ sites/8/2019/07/mh-care-pathway-231015.pdf (accessed 13 April 2024).
- 8. NHS England, NHS Improvement, National Collaborating Centre for Mental Health. *The Perinatal Mental Health Care Pathways*. 2018. URL: www. england.nhs.uk/wp-content/uploads/2018/05/perinatal-mental-health-care-pathway.pdf (accessed 13 April 2024).
- Conneely M, Packer KC, Bicknell S, Janković J, Sihre HK, McCabe R, et al. Exploring Black and South Asian women's experiences of help-seeking and engagement in perinatal mental health services in the UK. Front Psychiatry 2023;14:1119998. https://doi. org/10.3389/fpsyt.2023.1119998

- Moran E, Noonan M, Mohamad MM, O'Reilly P. Women's experiences of specialist perinatal mental health services: a qualitative evidence synthesis. Arch Womens Ment Health 2023;26:453–71. https://doi. org/10.1007/s00737-023-01338-9
- Jovanovic N, Packer K, Conneely M, Bicknell S, Copello A, McCabe R, *et al.* Barriers to accessing perinatal mental health services and suggestions for improvement – a qualitative study of women of Black and South Asian backgrounds. Accepted for publication in BJPSych Bulletin (15 July 2024).
- Jankovic J, Parsons J, Jovanović N, Berrisford G, Copello A, Fazil Q, Priebe S. Differences in access and utilisation of mental health services in the perinatal period for women from ethnic minorities – a population-based study. *BMC Med* 2020;**18**:1–12.
- Sartorius N, Jablensky A, Korten A, Ernberg G, Anker M, Cooper JE, Day R. Early manifestations and firstcontact incidence of schizophrenia in different cultures: a preliminary report on the initial evaluation phase of the WHO Collaborative Study on Determinants of Outcome of Severe Mental Disorders. *Psychol Med* 1986;**16**:909–28.
- 14. Volpe U, Mihai A, Jordanova V, Sartorius N. The pathways to mental healthcare worldwide: a systematic review. *Curr Opin Psychiatry* 2015;**28**:299–306.
- 15. GOV.UK Design System. *Equality Information*. 2022. URL: https://design-system.service.gov.uk/patterns/ equality-information/ (accessed 13 April 2024)
- 16. Race Disparity Unit. *List of Ethnicities*. 2022. URL: www.ethnicity-facts-figures.service.gov.uk/styleguide/ethnic-groups (accessed 13 April 2024).
- 17. Standards for Community Perinatal Mental Health Services. 6th edn. RCPsych; 2023. URL: www.rcpsych. ac.uk/docs/default-source/improving-care/ccqi/ quality-networks/perinatal/pqn-community-standards-6th-edition.pdf?sfvrsn=cf4efc12\_4 (accessed 13 April 2024).
- 18. Benjamini Y, Hochberg Y. Controlling the false discovery rate: a practical and powerful approach to multiple testing. *J R Stat Soc Series B Stat Methodol* 1995;**57**:289–300.
- 19. Raleigh V. The Health of People from Ethnic Minority Groups in England. The King's Fund; 2023. URL: www.kingsfund.org.uk/publications/health-people-ethnic-minority-groups-england#key-messages (accessed 13 April 2024).
- Bedaso A, Adams J, Peng W, Sibbritt D. Prevalence and determinants of low social support during pregnancy among Australian women: a community-based cross-sectional study. *Reprod Health* 2021;18:250. https://doi.org/10.1186/s12978-021-01210-y

- Kapadia D, Zhang J, Salway S, Nazroo J, Booth A, Villarroel-Williams N, *et al.* Ethnic inequalities in healthcare: a rapid evidence review NHS Race and Health Observatory. 2022. URL: www.nhsrho.org/ wp-content/uploads/2023/05/RHO-Rapid-Review-Final-Report\_Summary\_v.4.pdf (accessed 13 April 2024).
- 22. Watson H, Harrop D, Walton E, Young A, Soltani H. A systematic review of ethnic minority women's experiences of perinatal mental health conditions and services in Europe. *PLOS ONE* 2019;**14**:e0210587.
- 23. Edge D. Why are you cast down, o my soul? Exploring intersections of ethnicity, gender, depression, spirituality and implications for Black British Caribbean women's mental health. *Crit Public Health* 2013;**23**:39–48.
- 24. Williams E, Buck D, Babalola G, Maguire D. What are health inequalities? The King's Fund. 2023. URL: www. kingsfund.org.uk/insight-and-analysis/long-reads/ what-are-health-inequalities?gad\_source=1&gclid=CjwKCAjw5Ky1BhAgEiwA5jGujqx0v7escFy-J1Lc2niv9o\_ubv7WSD139pd1iZ41AdD204ZpgEaAvOBoCVBwQAvD\_BwE (accessed 13 April 2024).

#### Appendix 1 Study sample and patients with 'Mixed' and 'other' ethnicity

	All n = 228	Mixed n = 6 (2.6%)	<b>Other</b> n = 4 (1.8%)
Age, mean (SD)	30.3 (5.7)	3.8 (6.1)	35.5 (4.4)
Perinatal status, n (%)			
Pregnant	166 (72.8)	6 (100.0)	2 (50.0)
Postnatal	54 (23.7)	0 (0.0)	2 (50.0)
Preconception	8 (3.5)	0 (0.0)	0 (0.0)
Marital status, n (%)			
Single/divorced/living alone	36 (15.8)	1 (16.7)	1 (25.0)
Married/cohabitating/in relationship	189 (82.9)	5 (83.3)	3 (75.0)
Education, n (%)			
Primary + secondary	74 (32.5)	1 (16.7)	0 (0.0)
College + University	108 (47.4)	3 (50.0)	4 (100.0)
Employment, n (%)			
Employed	97 (42.5)	3 (50.0)	0 (0.0)
Unemployed	119 (52.2)	2 (33.3)	4 (100.0)
Number of children, Mdn (minimum-maximum, IQR)	Mdn = 1 (0-6, 2)	Mdn = 1 (0-3, 1.5)	Mdn = 1 (0-5, 3.75)
Involvement of children's social services, n (%)			
Yes	81 (35.5)	4 (66.7)	2 (50.0)
No	143 (62.7)	2 (33.3)	2 (50.0)

#### Note

Due to small sample sizes and non-normal distribution of ordinal/interval variables, the significance of differences in access to PMHS across ethnic groups was calculated only for Asian, Black and White patients (see *Table 2*).

### **Appendix 2** Complex patient pathways to PMHS (*n* = 84)

	Number of patients	Patient		1st service		2nd service		3rd service		4th service	_	5th service
Birmingham,	n = 20 (29%)	Patient	>	Midwife	>	Specialist midwife	>	PMHS				
n = 70 (83.3%)	n = 10 (14%)	Patient	>	CMHT	>	Midwife	>	PMHS				
	n = 4 (5.7%)	Patient	>	Midwife	>	CMHT	>	PMHS				
	n = 3 (4.3%)	Patient	>	GP	>	Talking therapies	>	PMHS				
	n = 2 (2.8%)	Patient	>	CMHT	>	Specialist midwife	>	PMHS				
	n = 2 (2.8%)	Patient	>	Obstetrician	>	Specialist midwife	>	PMHS				
	n = 2 (2.8%)	Patient	>	GP	>	GP	>	PMHS				
	n = 2 (2.8%)	Patient	>	GP	>	CMHT	>	PMHS				
	n = 2 (2.8%)	Patient	>	Midwife	>	GP	>	Specialist midwife	>	PMHS		
	n = 1 (1.4%)	Patient	>	Obstetrician	>	Specialist midwife	>	GP	>	PMHS		
	n = 1 (1.4%)	Patient	>	Obstetrician	>	Specialist midwife	>	Liaison psychiatry	>	PMHS		
	n = 1 (1.4%)	Patient	>	CMHT	>	Crisis team	>	Midwife	>	PMHS		
	n = 1 (1.4%)	Patient	>	CMHT	>	Obstetrician	>	PMHS	>			
	n = 1 (1.4%)	Patient	>	CMHT	>	Midwife	>	Specialist midwife	>	PMHS		
	n = 1 (1.4%)	Patient	>	GP	>	Specialist midwife	>	Liaison psychiatry	>	PMHS		
	n = 1 (1.4%)	Patient	>	GP	>	Crisis team	>	PMHS				
	n = 1 (1.4%)	Patient	>	GP	>	CMHT	>	Midwife	>	Specialist midwife	>	PMHS
	n = 1 (1.4%)	Patient	>	GP	>	Specialist midwife	>	PMHS				
	n = 1 (1.4%)	Patient	>	Midwife	>	GP	>	PMHS				
	n = 1 (1.4%)	Patient	>	Midwife	>	GP	>	PMHS	>	Liaison psychiatry	>	PMHS
	n = 1 (1.4%)	Patient	>	Midwife	>	GP	>	CMHT	>	PMHS		
	n = 1 (1.4%)	Patient	>	Midwife	>	Specialist midwife	>	GP	>	GP	>	PMHS
	n = 1 (1.4%)	Patient	>	Midwife	>	GP	>	GP	>	PMHS		
	n = 1 (1.4%)	Patient	>	Midwife	>	Crisis team	>	PMHS				
	n = 1 (1.4%)	Patient	>	Midwife	>	GP	>	Obstetrician	>	Specialist midwife	>	PMHS
	n = 1 (1.4%)	Patient	>	Midwife	>	Obstetrician	>	GP	>	PMHS		
	n = 1 (1.4%)	Patient	>	Midwife	>	CMHT	>	Specialist midwife	>	PMHS		
	n = 1 (1.4%)	Patient	>	Midwife	>	Midwife	>	PMHS				
	n = 1 (1.4%)	Patient	>	Liaison psychiatry	>	Specialist midwife	>	PMHS				
	n = 1 (1.4%)	Patient	>	Psychiatric ward	>	Liaison psychiatry	>	Crisis team	>	CMHT	>	PMHS
	n = 1 (1.4%)	Patient	>	A&E	>	Liaison psychiatry	>	Crisis team	>	PMHS		
	n = 1 (1.4%)	Patient	>	Medical ward	>	Liaison psychiatry	>	Specialist midwife	>	PMHS		

16

	Number of patients	Patient		1st service		2nd service		3rd service		4th service		5th service
London, <i>n</i> = 14	n = 1 (7.1%)	Patient	>	GP	>	Midwife	>	PMHS				
(16.7%)	n = 1 (7.1%)	Patient	>	GP	>	Crisis team	>	PMHS				
	n = 1 (7.1%)	Patient	>	GP	>	CMHT	>	PMHS				
	n = 1 (7.1%)	Patient	>	GP	>	Liaison psychiatry	>	CMHT	>	PMHS		
	n = 1 (7.1%)	Patient	>	GP	>	Talking therapies	>	PMHS				
	n = 1 (7.1%)	Patient	>	Midwife	>	Specialist midwife	>	PMHS				
	n = 1 (7.1%)	Patient	>	Midwife	>	PMHS	>	PMHS				
	n = 1 (7.1%)	Patient	>	Fertility clinic	>	GP	>	PMHS				
	n = 1 (7.1%)	Patient	>	Talking therapies	>	CMHT	>	PMHS				
	n = 1 (7.1%)	Patient	>	Postnatal ward	>	Liaison psychiatry	>	PMHS				
	n = 1 (7.1%)	Patient	>	A&E	>	Liaison psychiatry	>	PMHS				
	n = 1 (7.1%)	Patient	>	A&E	>	Psychiatric ward	>	PMHS				
	n = 1 (7.1%)	Patient	>	Liaison psychiatry	>	Crisis team	>	СМНТ	>	PMHS		
	n = 1 (7.1%)	Patient	>	Crisis line	>	A&E	>	Liaison psychiatry	>	A&E	>	PMHS

CMHT, Community Mental Health Team; Crisis team, crisis resolution and home treatment team; GP, general practitioner.

DOI: 10.3310/YTRK6337